Author’s response to reviews

Title: Acceptability, effectiveness, and cost-effectiveness of internet-based exposure treatment for irritable bowel syndrome in a clinical sample: a randomized controlled trial

Authors:

Brjánn Ljótsson (brjann.ljotsson@ki.se)
Gerhard Andersson (gerhard.andersson@liu.se)
Erik Andersson (erik.m.andersson@ki.se)
Erik Hedman (erik.hedman.2@ki.se)
Perjohan Lindfors (perjohan.lindfors@aleris.se)
Sergej Andréewitch (andreewitch@gmail.com)
Christian Rück (christian.ruck@ki.se)
Nils Lindefors (nils.lindefors@ki.se)

Version: 7 Date: 11 October 2011

Author’s response to reviews:

p. 20. Last paragraph:

"The acceptability of ICBT, as measured by the proportions of patients ICBT staying in the study (77%) and treatment completers (43%)..."

has been changed to

"The acceptability of ICBT, as measured by the proportions of patients in the ICBT group that stayed in the study (77%) and completed treatment (43%)..."